Author's response to reviews

Title: Application of simple swallowing provocation test with fiberoptic endoscopic evaluation of swallowing in a cross-sectional study

Authors:

Chiharu Tejima (chiharu-tejima@umin.ac.jp)
Takeshi Kikutani (kikutani@tokyo.ndu.ac.jp)
Noriaki Takahashi (high-bri@tokyo.ndu.ac.jp)
Fumiyo Tamura (fumita@tokyo.ndu.ac.jp)
Mitsuyoshi Yoshida (mitsu@hiroshima-u.ac.jp)

Version: 3 Date: 23 February 2015

Author's response to reviews: see over
Feb. 10, 2015

Dr. Giulia Mangiameli  
Editor in Chief, BMC Geriatrics  
United Kingdom

Ref: MS 6052788211560170
Application of simple swallowing provocation test with fiberoptic endoscopic evaluation of swallowing in a cross-sectional study
Chiharu Tejima, Takeshi Kikutani, Noriaki Takahashi, Fumiyo Tamura and Mitsuyoshi Yoshida

Dear Dr. Mangiameli

Thank you for your e-mail of February 3, 2015, regarding our manuscript, “Application of simple swallowing provocation test with fiberoptic endoscopic evaluation of swallowing in a cross-sectional study”, and the valuable comments of the two reviewers. I attach here our revised manuscript, as well as a point-by-point response to the reviewers’ comments.

We feel that the revised manuscript is a suitable response to the comments, and is significantly improved over the initial submission. We trust that it is now suitable for publication in BMC Geriatrics. Thank you in advance for your kind consideration of this paper.

Sincerely yours,

Chiharu Tejima, DDS  
Prof. Takeshi Kikutani, DDS, PhD,  
Division of Clinical Oral Rehabilitation,  
The Nippon Dental University, Graduate School of Life Dentistry of Tokyo  
1-9-20, Fujimi, Chiyoda-ku, Tokyo, 102-0071, JAPAN